Author’s response to reviews

Title: Functional Independence Measure score is associated with mortality in critically ill elderly patients admitted to an intermediate care unit.

Authors:

Alexia D'Andrea (dandrea.alexia@gmail.com)
Damien Le Peillet (Damien.lePeillet@hcuge.ch)
Thomas Fassier (Thomas.Fassier@hcuge.ch)
Virginie Prendki (Virginie.Prendki@hcuge.ch)
Veronique Trombert (Veronique.Trombert@hcuge.ch)
Jean-Luc Reny (Jean-Luc.Reny@hcuge.ch)
xavier roux (xavier.roux@hcuge.ch)

Version: 3 Date: 20 Aug 2020

Author’s response to reviews:

Dear editor,

You'll find our comment below for each question. Best regards.

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors VP, VT, AND TF do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

Changes were made into the text with some precisions about each autor contribution.

2. Please reformat the Authors’ Contributions section so that it is presented as a paragraph, and not as a list. Please also ensure that in this section, the authors’ full initials are provided [first name
initial, last name initial], in order to correctly correspond to the list of authors in the submissions system for Alexia D'Andrea and Damien Le Peillet.

Done into the text, as requested.

3. Ethics Approval and Consent to Participate
Please clarify whether any administrative permissions were required to access and use the medical records described in your study. If so, please state who granted permission in the “Ethics approval and consent to participate” section.

Precision put into the text.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.